Author’s response to reviews

Title: Development and Validation of an Algorithm to Recalibrate Mental Models and Reduce Diagnostic Errors Associated with Catheter-Associated Bacteriuria

Authors:

Barbara W Trautner (Trautner@bcm.edu)
Rupal D Bhimani (Rupal.Bhimani@va.gov)
Amber B Amspoker (Amspoker@bcm.edu)
Sylvia J Hysong (hysong@bcm.edu)
Armandina Garza (Armandina.Garza@va.gov)
Peter A Kelly (pakelly@tulane.edu)
Velma L Payne (vpayne@bcm.edu)
Aanand D Naik (anaik@bcm.edu)

Version: 2 Date: 21 August 2012

Author’s response to reviews:

Dear Review Committee,

We appreciate the opportunity to revise the manuscript and address comments from the editors. We have addressed the editorial office’s recommendation that we should include an Ethics Statement in the Methods Section of the manuscript. We have added the following sentence to the end of the first paragraph in the methods section: "This research was conducted with the approval of the Baylor College of Medicine Internal Review Board (protocol H #24180)." Thank you for your feedback and consideration of our manuscript.

Sincerely,

Aanand Naik